Author's response to reviews

Title: Estimating the lifetime economic burden of stroke according to the age of onset in South Korea: a cost of illness study

Authors:

Hye-Young Kang (hykang2@yonsei.ac.kr)
Seung-Ji Lim (sjlim@yuhs.ac)
Hae Sun Suh (haesun.suh@yonsei.ac.kr)
Danny Liew (dliew@medstv.unimelb.edu.au)

Version: 6 Date: 11 August 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

The functioning link to "http://www.index.go.kr/egams/stts/jsp/potal/stts/PO_STTS_IdxMain.jsp?idx_cd=1063" works when I click it. However, since this link doesn't work for you, I changed it to "https://www.index.go.kr/egams/stts/jsp/potal/stts/PO_STTS_IdxMain.jsp?idx_cd=1063&bbs=INDX_001" which is the same webpage.

If this link does not work again, it might be because the webpage is written in Korean. Please let me know if the new link still does not function.

For minor revisions (Figures and Typography), I would like you to make these changes.

Sincerely,

Hae Sun Suh